Author's response to reviews

Title: Association between Helicobacter pylori cagA-related genes and clinical outcomes in Colombia and Japan

Authors:

Masahide Watada (m08d9036@oita-u.ac.jp)
Seiji Shiota (sshiota@oita-u.ac.jp)
Osamu Matsunari (narry@oita-u.ac.jp)
Rumiko Suzuki (rusuzuki@oita-u.ac.jp)
Kazunari Murakami (murakam@oita-u.ac.jp)
Toshio Fujioka (fujioka@oita-u.ac.jp)
Yoshio Yamaoka (yyamaoka@bcm.tmc.edu)

Version: 4 Date: 2 December 2011

Author's response to reviews: see over
Dear Dr. Damian Hussey,

Editor, BMC Gastroenterology:

December 2, 2011

Thank you for your letter. We were pleased to receive a positive evaluation of our manuscript. The manuscript was modified according to Editor’s comments.

The change made during the revision is printed in blue letters in the revised manuscript.

We hope that the revised manuscript is now acceptable for publication in BMC Gastroenterology.

Yours sincerely,

Yoshio Yamaoka, MD, PhD

Department of Environmental and Preventive Medicine, Oita University Faculty of Medicine, 1-1 Idaigaoka, Hasama-machi, Yufu-City, Oita 879-5593, Japan
Tel: +81-97-586-5740
Fax: +81-97-586-5749
E-mail: vyamaoka@oita-u.ac.jp
ITEMIZED RESPONSES

Responses to comments raised by Editor

On pg 2, line 10 in the methods section of the abstract, 323 strains should be corrected to 343 strains.

Response:
Thank you for your comment. We really appreciate your point. We corrected the number from 323 to 343 in page 2, line 10 in the revised manuscript.